Author's response to reviews

Title: Reduced Crying in Term Infants Fed High Beta-Palmitate Formula: A Double-Blind Randomized Clinical Trial

Authors:

Ita Litmanovitz (litmani@clalit.org.il)
Fabiana Bar-Yoseph (fabiana@ENZYMOTEC.COM)
Yael Lifshitz (Yaell@ENZYMOTEC.COM)
Keren Davidson (davidson.keren@gmail.com)
Alon Eliakim (Eliakim.Alon@clalit.org.il)
Rivka regev (regevr@clalit.org.il)
Dan Nemet (Dan.Nemet@clalit.org.il)

Version: 10 Date: 9 June 2014

Author's response to reviews: see over
The Editor, BMC Pediatrics

RE: MS: 8173406141172559 – R3

Reduced Crying in Term Infants Fed High Beta-Palmitate Formula: A Double-Blind, Randomized, Clinical Trial

Dear Editor,

Thank you for your e-mail dated 06 June 2014, regarding our MS “Reduced Crying in Term Infants Fed High Beta-Palmitate Formula: A Double-Blind, Randomized, Clinical Trial” notifying us that the review of our MS was completed and that it is suitable for publication in the journal, pending editorial revisions.

As suggested, we have completed the editorial revision, and the revised MS is now attached.

We appreciate the opportunity to submit this research and hope that it will now be found suitable for publication in BMC Pediatrics.

Sincerely yours,

Ita Litmanovitz, MD

Corresponding Author
Editorial Comments

Abstract

Please include the date that your trial was registered, along with the TRN number at the end of your abstract. Please note that the abstract on the submission system and that in the main body of the manuscript should be identical.

The trial registration date and number are now added to the MS.

We have changed the abstract so that now the abstract on the submission and the MS are identical.

Additional formatting request:

1. Title page: Please remove the suffixes from the title page

Done

2. CONSORT guidelines: a consort statement was uploaded to the website

3. Ethics: A statement concerning parental or guardian consent appears in the manuscript